Author’s response to reviews

Title: Predictors for gram-negative monomicrobial necrotizing fasciitis in southern Taiwan

Authors:

Tsung-Yu Huang (r12045@cgmh.org.tw)
Kuo-Ti Peng (3497@cgmh.org.tw)
Cheng-Ting Hsiao (qcth3160@cgmh.org.tw)
Wen-Chih Fann (patrickfann@cgmh.org.tw)
Yao-Hung Tsai (orma2244@cgmh.org.tw)
Yen-Yao Li (orthoyao@cgmh.org.tw)
Chien-Hui Hung (hungc01@mail.cgu.edu.tw)
Fang-Yi Chuang (daisy5670@cgmh.org.tw)
Wei-Hsiu Hsu (7572@cgmh.org.tw)

Version: 2 Date: 08 Jan 2020

Author’s response to reviews:

Assistant Editor Comments:

1. Consent to participate

In your "ethics approval and consent to participate" section of your declarations, please clarify whether the consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

2. Role of funder

In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. If they had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript please clearly state this.

3. Rename Materials and Methods
Please rename Materials and Methods to 'Methods'.

4. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.

Author response:

1. Thank you for your kind reminder. We had added ethics approval and consent to participate in Methods (Line 7 in Page 8).

2. Thank you for your kind reminder. We had added role of the funding in Declarations (Line 22 in Page 16).

3. Thank you for your kind reminder. We had rename Materials and Methods to 'Methods'.

4. Thank you for your kind reminder.

GB Firth, MBBCh, FCS (Orth). MMed (Orth) (Reviewer 1):

The revision has substantially improved the quality of this article.

Congratulations to all the authors on the work done.

All of my comments and queries were addressed adequately.

Author response:

Thank you for your recognition of the importance of this study.